\documentclass[10pt,twocolumn]{article}
\usepackage{times}
\usepackage{parskip}
\usepackage{geometry}
\usepackage{url}
\usepackage{color}
\usepackage{mdwlist}
\usepackage{graphicx}

\geometry{margin=1in}

\definecolor{grey}{rgb}{0.5, 0.5, 0.5}

\title{%
  The Wossname Thingamajig\\
  {\large\textcolor{grey}{iPhone Programming Praktikum WS 2009/2010}}
}
\author{%
  Alice Schmidt\\cand. M.Sc.\\FU Berlin\\Student-ID 123456\\
  \url{alice@inf.fu-berlin.de}
  \and
  Bob Muller\\cand. Dipl.-Inform.\\FU Berlin\\Student-ID 456789\\
  \url{bob@inf.fu-berlin.de}
}
\date{\today}

\begin{document}
\maketitle

\section{Overview}

This section describes the goals and basic aspects of the application that
the team intends to develop.

\section{Components}

This sections gives a break down of the components that the application will
have or need.  For example:
\begin{itemize*}
\item The iPhone application itself
\item A matching server component running on another computer.
\item A data preparation tool running on another computer.
\end{itemize*}

\section{Frameworks and Libraries}

This section summarizes the iPhone OS Frameworks, external services and
external libraries that will likely be required for the application.  This
information is needed to determine the feasibility of the project.  For
example, an implementation of \emph{Mental Poker} may be feasible within the
scope of the Praktikum if the team is aware of \texttt{libTMCG} but will
hardly be feasible otherwise.

\section{Features}

\subsection{Minimal Feature Set}

This section lists the application features required for a full score in the
\emph{architecture} grading category.

\subsection{Targeted Feature Set}

This section lists the application features required on top of the minimal
features for a full score in the \emph{feature completion} grading category.

\section{References}

This section lists references to publications of algorithms that will be
used in the project.  Only algorithms that have a special significance to
the project shall be referenced.

\end{document}
